Author's response to reviews

Title: Cardiovascular Health in Anxiety or Mood Problems Study (CHAMPS): protocol for a feasibility study for the transdiagnostic treatment of emotional disorders among cardiovascular patients

Authors:

Phillip Tully (phillip.tully@adelaide.edu.au)
Deborah A Turnbull (deborah.turnbull@adelaide.edu.au)
John D Horowitz (john.horowitz@adelaide.edu.au)
John F Beltrame (john.beltrame@adelaide.edu.au)
Terina Selkow (Terina.Selkow@health.sa.gov.au)
Elizabeth Markwick (Elizabeth.Markwick@health.sa.gov.au)
Bernhard T Baume (bernhard.baume@adelaide.edu.au)
Shannon Shannon Sauer-Zavala (shannonezavala@gmail.com)
Harald Baumeister (harald.baumeister@psychologie.uni-freiburg.de)
Gary A Wittert (gary.wittert@adelaide.edu.au)

Version: 2 Date: 15 June 2015

Author's response to reviews:

Revised filed per journal guidelines attached